Author's response to reviews

Title: A diagnostic approach for defining Idiopathic Remitting Diabetes: a retrospective cohort study

Authors:

Tarig Babiker (t.babiker@exeter.ac.uk)
Ali Chakera (alichakera@nhs.net)
Maggie Shepherd (M.H.Shepherd@exeter.ac.uk)
Andrew T Hattersley (A.T.Hattersley@exeter.ac.uk)

Version: 2 Date: 27 September 2013

Author's response to reviews:

An ethics statement has been included which states:

This study was not deemed to require REC approval under GAfREC guidelines as it involved research undertaken by staff within a care team using information previously collected in the course of care for their own patients or clients, which was pseudonymised in conducting the research.